Author’s response to reviews

Title: Relationship between self-efficacy, self-care behaviour and glycaemic control among patients with type 2 diabetes mellitus in the Malaysian primary care setting

Authors:
Zahirah Tharek (ztharek@yahoo.com)
Anis Safura Ramli (rossanis_yuzadi@yahoo.co.uk)
David Whitford (dwhitford@rcsi.ie)
Zaliha Ismail (drzie65@gmail.com)
Maryam Mohd-Zulkifli (meccha1607@yahoo.com)
Siti Khuzaimah Ahmad-Sharoni (sitik123@salam.uitm.edu.my)
Asrul Shafie (aakmal@usm.my)
Thevaraaajan Jayaraman (theva.jayaraman@gmail.com)

Version: 1 Date: 29 May 2017

Author’s response to reviews:

Editors-in-Chief
BMC Family Practice

29th May 2017

Dear Editor,

Thank you for your kind consideration that our manuscript is potentially acceptable for publication in BMC Family Practice.
On behalf of all the authors, I wish to resubmit the revised version of this manuscript entitled ‘Relationship between self-efficacy, self-care behaviour and glycaemic control among patients with type 2 diabetes mellitus in the Malaysian primary care setting’. All changes are highlighted in yellow in the revised manuscript.

Please find attached as a supplementary file: the point-by-point detailed responses to each of the reviewer 1 and 2 comments, describing the amendments which have been made to the manuscript text and where these can be found.

All authors have read and approved this revised version. We confirm that this work is our original work. The manuscript has not been published or submitted for publication elsewhere.

The authors declare that we have no conflict of interest.

Thank you for your kind consideration.

Yours sincerely,

Associate Professor Dr. Anis Safura Ramli,
Consultant Primary Care Physician & Research Fellow,
Institute for Pathology, Laboratory and Forensic Medicine (I-PPerForM),
Universiti Teknologi MARA (UiTM), Sungai Buloh Campus,
Jalan Hospital, 47000 Sungai Buloh, Selangor, Malaysia.